Dear Editor:

Attached is the revised manuscript entitled "Reliability of Measures of Impairments Associated With Patellofemoral Pain Syndrome". We have made the following formatting changes requested:

- Title & Abstract - Are now identical in the manuscript and in the submission system.
- Author list - The list of authors in the manuscript is now written exactly as they are in the submission system.
- Highlighting/Tracking - We removed all highlighting/tracking from the manuscript text.
- References - We placed the references before the full stop.
- Introduction - this section was renamed 'Background'.
- Section headings - Are all in sentence case. We also removed unnecessary capitals from Table 2.
- Discussion- We removed the unnecessary information about the SPSS.
- Author's contributions - Was changed to Authors' contributions.

This revised manuscript has been read and agreed by all authors.

Thank you for your consideration.

Sincerely,

Sara Piva, PT, PhD, OCS, FAAOMPT
Assistant Professor
Department of Physical Therapy
University of Pittsburgh
6035 Forbes Tower
Pittsburgh, PA 15260 (USA)
Phone: (412) 383-8712
Fax: (412) 383-6629